Author's response to reviews

Title: Prehospital score for acute disease: a community-based observational study in Japan

Authors:

Yasuhiro Toyoda (ytoyoda@pbhel.med.osaka-u.ac.jp)
Yoshio Matsuo (kch-25_g9q9@kch.city.kishiwada.osaka.jp)
Hiroyuki Tanaka (ertanaka@tokyo-med.ac.jp)
Hidekazu Fujiwara (kishiwada-kyukyu@tvk.zaq.ne.jp)
Toshio Takatorige (takatorige@pnhel.med.osaka-u.ac.jp)
Hiroyasu Iso (iso@pbhel.med.osaka-u.ac.jp)

Version: 3 Date: 25 August 2007

Author's response to reviews:

Dr. Melissa Norton
Editorial Director, BMC Emergency Medicine

Dear Dr. Melissa Norton

Enclosed you will find the revised manuscript, "Prehospital score for acute disease: a community-based observational study in Japan" as a research article to BMC Emergency Medicine.

We revised the manuscript based on the reviewer's comments. We also described "Author's comments".

All authors have read and approved re-submission of the manuscript, and that material in the manuscript has not been published and is not being considered for publication elsewhere in whole or in part in any language.

All authors don't have any potential conflicts of interest or copyright constraints.

Sincerely,

Professor Hiroyasu Iso, MD
Public Health, Department of Social and Environmental Medicine, Graduate School of Medicine, Osaka University, Yamadaoka Suita-shi, Osaka 565-0871, Japan.
Phone: +81 (6) 6879-3911 Fax: +81 (6) 6879-3919
E-mail: iso@pbhel.med.osaka-u.ac.jp